Author's response to reviews

Title: Western Australia Food Security Project

Authors:

    Alexandra McManus (A.Mcmanus@curtin.edu.au)
    Bruce Maycock (B.Maycock@curtin.edu.au)

Version: 4 Date: 3 August 2007

Author's response to reviews:

To the Editor

The minor typographical errors have been amended as requested.

One reviewer has requested a copy of the survey instrument be included. As this would make the article rather long, we have allowed the opportunity for anyone interested to contact the lead author for a copy of the instrument.

Delicatessans has been defined within the text and the fact that food outlet owner/operators were identified from council records has also been included in text.

This accounts for all reviewers requests.

Kind regards
Alexandra